Author's response to reviews

Title: Zinc sulfate as an adjunct to methylphenidate for the treatment of attention deficit hyperactivity disorder in children: A double blind and randomized trial [ISRCTN64132371]

Authors:

   Dr Shahin Akhondzadeh (s.akhond@neda.net)
   Prof Mohammad-Reza Mohammadi (mrmohammadi@yahoo.com)
   Mojghan Khademi (a3057@sina.tums.ac.ir)

Version: 4 Date: 18 Mar 2004

Dear Madam/Sir

The figures were redesigned as you requested.

Sincerely,
Shahin Akhondzadeh Ph.D.